

## **Author's response to reviews**

**Title:** Prospective study of urinary tract infection surveillance after kidney transplantation

### **Authors:**

Roberto Ribera-Sanchez ([cerdea321@yahoo.com.mx](mailto:cerdea321@yahoo.com.mx))  
Dolores Delgado-Ochoa ([lolgera@hotmail.com](mailto:lolgera@hotmail.com))  
Rocio Flores-Paz ([rflorespaz@yahoo.com.mx](mailto:rflorespaz@yahoo.com.mx))  
Elvia García-Jimenez ([elviagen@yahoo.com.ms](mailto:elviagen@yahoo.com.ms))  
Ramon Espinoza-Hernandez ([espinohdZR2003@yahoo.com.mx](mailto:espinohdZR2003@yahoo.com.mx))  
Andres Bazan-Burgosa ([bazandr@hotmail.com](mailto:bazandr@hotmail.com))  
Myriam Arriaga-Alba ([arriaga\\_alba@yahoo.com](mailto:arriaga_alba@yahoo.com))

**Version:** 2 **Date:** 27 July 2009

### **Author's response to reviews:**

Dear Sirs:

As you have asked us, we have sent the paper "Prospective study of urinary tract infections, surveillance after renal transplant surgery".- has been sent to (<http://www.biomedcentral.com/my>)., in order to have the paper corrected.

The original title has been changed as reviewer's suggestions to: Prospective study of urinary tract infection surveillance after kidney transplantation.

The text is in fact the same, but they have done some English redaction corrections. Some mistakes in the references section have been corrected.

Table 3, has a new column explaining the percentage of patients with UTI or without UTI (Urinary Tract Infections) who have rejected its renal allograft. It was added in order to give the lector a more clear knowledge of the risk of having UTI after renal kidney transplantation.

Yours faithfully:

Myriam Arriaga\_Alba.